Author's response to reviews

Title: Compensatory quadrant-hyperhidrosis after contralateral intrathoracic surgery: a case report

Authors:

Stefan Brodoehl (stefan.brodoehl@med.uni-jena.de)
Albrecht Guenther (albrecht.guenther@med.uni-jena.de)
Otto OW Witte (otto.witte@med.uni-jena.de)

Version: 2 Date: 28 August 2012

Author's response to reviews: see over
Submission of a Case report in Journal of Medical Case Reports

Dear Editor,

We hereby submit our manuscript entitled “Compensatory quadrant-hyperhidrosis after contralateral intrathoracic surgery” for publication as a Case report in Journal of Medical Case Reports.

Herein we present the case of a 67-year-old man suffering from a hyperhidrosis of the right upper body quadrant (including face, arm, and chest) after intrathoracic surgery of a left sided pleural lipoma. In addition to a short pathophysiological summary, we describe a successful therapeutic strategy using botulinum toxin type A and present impressive clinical photography showing significant differences in sweating and temperature between the right and left upper body.

We believe that this manuscript is of interest to the broad clinical and scientific readership of Journal of Medical Case Reports.

We assure that this manuscript is not under simultaneous consideration by any other publication.

Yours sincerely,

Stefan Brodoehl, M.D.  Albrecht Guenther, Otto W. Witte, M.D.¹